ACCU ADVANCING EARTH AND SPACE SCIENCE

Geochemistry, Geophysics, Geosystems[°]

RESEARCH ARTICLE

10.1029/2022GC010600

Key Points:

- Coral skeletal Cd/Ca ratios in monthly resolution were first reported
- Seawater mixing process and sediments resuspension caused by winter monsoon are responsible for Cd seasonal variations in surface seawater
- Coral skeletal Cd/Ca ratios have potential to be a novel proxy for winter monsoon

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

W. Jiang and K. Yu, jianwe@gxu.edu.cn; kefuyu@scsio.ac.cn

Citation:

Sun, Y., Jiang, W., Yu, K., Xu, S., Feng, C., Xie, S., & Wei, C. (2022). High-resolution coral records of cadmium in surface seawater: Biogeochemical cycling and a novel proxy for winter monsoon. *Geochemistry, Geophysics, Geosystems,* 23, e2022GC010600. https:// doi.org/10.1029/2022GC010600

Received 29 JUN 2022 Accepted 10 OCT 2022

Author Contributions:

Conceptualization: Wei Jiang Data curation: Wei Jiang, Shendong Xu Formal analysis: Yinan Sun, Wei Jiang Funding acquisition: Kefu Yu Investigation: Yinan Sun, Shendong Xu, Chunmei Feng, Sirong Xie Methodology: Wei Jiang, Chaoshuai Wei Project Administration: Kefu Yu Resources: Kefu Yu, Shendong Xu, Chunmei Feng, Sirong Xie Software: Yinan Sun, Wei Jiang Supervision: Wei Jiang, Kefu Yu Validation: Chaoshuai Wei

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

High-Resolution Coral Records of Cadmium in Surface Seawater: Biogeochemical Cycling and a Novel Proxy for Winter Monsoon

Yinan Sun¹, Wei Jiang^{1,2}, Kefu Yu^{1,2}, Shendong Xu^{1,2}, Chunmei Feng¹, Sirong Xie^{1,3}, and Chaoshuai Wei¹

¹Guangxi Laboratory on the Study of Coral Reefs in the South China Sea, Coral Reef Research Center of China, School of Marine Sciences, Guangxi University, Nanning, China, ²Southern Marine Science and Engineering Guangdong Laboratory (Zhuhai), Zhuhai, China, ³School of Resources, Environment and Materials, Guangxi University, Nanning, China

Abstract Cadmium (Cd) geochemical cycle plays a significant role in the composition and function of the marine ecosystem. Skeletal cadmium-to-calcium (Cd/Ca) ratios in hermatypic corals have been applied to reconstruct the historical changes of oceanic and climatic processes, yet there was no systematic evaluation of this tracer's natural variability in high resolution over time. Here, we reported a coral skeletal Cd/Ca record in monthly resolution from 1999 to 2008 CE and reconstructed the history of Cd contents in surface seawater in the northern South China Sea. A significant seasonal variation (higher in the winter but lower in the summer) of Cd contents in surface seawater can be identified. We found that the seasonal variations in coral skeletal Cd/Ca ratios exhibited significant trends coupled with the surface wind speeds, indicating that strong winds had likely driven the vertical seawater mixing process and then induced the process of sediment remobilization on the shelf, which significantly increased Cd contents in surface seawater. The reduction in Cd contents in surface seawater due to biological processes might be masked by the impacts of surface winds. Importantly, we also observed that coral skeletal Cd/Ca records in the winter showed significant correlations with the winter monsoon index, highlighting the possibility as a new proxy of winter monsoon in the non-upwelling shelf environments.

Plain Language Summary Cadmium (Cd) is a nutrient element absorbed by phytoplankton in surface seawater, showing a distribution of low surface content and high bottom content. In order to determine the seasonal variations of Cd content in surface seawater in the northern South China Sea, we analyzed the coral skeletal Cd/Ca ratios in monthly resolution (1999–2008 CE) and finally concluded that the process of sediment resuspension and vertical mixing of seawater driven by winter monsoons is the main reason. The reduction of Cd contents in surface seawater due to biological processes was masked by the impacts of surface winds. In the past, Cd content in seawater/coral has been used to track the history of upwelling and El Niño Southern Oscillation, but the potential indicative on winter monsoon has not been reported so far. Considering that the trend between coral skeletal Cd/Ca ratios and average wind speeds showed a great agreement, we suggested that coral skeletal Cd/Ca ratios in the winter might be a potential proxy of winter monsoon.

1. Introduction

Cadmium (Cd) is a toxic trace metal element, which can be transferred or accumulated through the food chain, and is one of the elements that must be measured in marine monitoring (Jurado-González et al., 2003). The behavior and distribution of Cd in the seawater are similar to that of macro-nutrient, being biologically depleted in surface seawater (Xu et al., 2008) and enriched at depth through dissolution of sinking organic matter (Boyle et al., 1976; Bruland et al., 1978), forming a scavenged depth profile of nutrient types (Grottoli et al., 2013). Cd is also considered as a micro-nutrient associated with a carbonic anhydrase that is ubiquitous in the bacteria, plants and animal kingdoms (Lionetto et al., 2005). Some researches suggested that Cd was an essential trace metal element in phytoplankton (Lane & Morel, 2000; Lane et al., 2005), but others suggested that Cd was non-specifically exploited by phytoplankton and stored intracellularly to avoid toxicity (Horner et al., 2013). Whichever mechanism may explain the nutrient-like behavior of Cd, its strong correlation with phosphate reflects its involvement in biological cycling processes (Delgadillo-Hinojosa et al., 2015). However, the geochemical



Visualization: Wei Jiang Writing – original draft: Yinan Sun Writing – review & editing: Yinan Sun, Wei Jiang, Kefu Yu behavior of Cd in surface seawater is still poorly resolved, despite its potentially critical impacts on the nearshore biosphere.

In the modern ocean, the primary source of Cd is land runoff in particulate forms (Libes, 2009; Wang et al., 2010). As river water moves downstream into estuaries, a large amount of particulate matters enter the ocean, most of which is decomposed and bound in surface seawater by various organisms (Collier & Edmond, 1984; Martin & Knauer, 1973), while a small portion is adsorbed by particulate matters and settles on the seafloor along with other organic debris, which also remains in a state of continued decomposition during the sinking process (Boyle, 1988). As a nutrient-like trace metal element, Cd below the light-transmitting layer increases rapidly with depth and reaches a maximum near the thermocline (\sim 1,000 m) (Libes, 2009). After that, it no longer changes significantly with the water depth and remains constant basically (Feng & Tian, 2018; Libes, 2009). The retention time of Cd in seawater is ~300,000 years, much longer than the sufficiently mixing time of seawater, so that Cd from natural and anthropogenic sources is mainly deposited in the bottom sediments (Zhang et al., 2019). The Cd contents in sediments vary in the range of $0-8.6 \ \mu g/g$ (Yang et al., 2021). The ocean has a scavenging effect, usually in the form of suspended sediments adsorption (Turner et al., 2008), metal-sulfide formation (Jiann et al., 2005) and biological ingestion (Liu et al., 2012). Cd may be re-released into the surface seawater by upwelling and ocean currents (Delgadillo-Hinojosa et al., 2015), as a "new" source (Souri et al., 2020). Little environmental concern is warranted for the remobilization of Cd (Choi et al., 2006), and the desorption of Cd from sediments by resuspension in seawater is greater than for any other heavy metals, controlled by the redox potential (Simpson, 1978). In addition, Cd content is low in surface seawater of the tropical and subtropical regions, but high in middle and deep water of the Southern Ocean (Feng & Tian, 2018). This feature is mainly related to the mixing of water masses, which also indicates that Cd is sensitive to oceanic processes such as vertical mixing and upwelling (Carriquiry and Villaescusa, 2010). Therefore, the variation of Cd contents in seawater is often used as an indicator of ocean current changes.

Aragonite skeletons of corals are ideal geochemical indicators, providing the marine environment and meteorological information (Saha et al., 2016), and the Metal/Ca ratios in coral skeletons with the properties of high-resolution and continuous records are unique archives recording temporal variations of trace metals in surface seawater (Saha et al., 2016). The coral cadmium/calcium (Cd/Ca) is generally regarded as an indicator of upwelling (Reuer et al., 2003). Shen et al. (1987) believed that Cd in reef corals is a sensitive tracer of natural and anthropogenic disturbances to the surface seawater in history, and can be used to evaluate the onset/termination timing patterns of El Niño Southern Oscillation (ENSO). Pretet et al. (2014) found that in laboratory experiments Cd/Ca radios and salinity revealed a positive correlation for each coral genus. Carriquiry and Villaescusa (2010) found that the interannual variation of coral skeletal Cd/Ca radios in the eastern North Atlantic is related to ENSO, and suggested that Cd/Ca radios in biogenic carbonates could be used as a tracer of increasing seawater stratification and weakening trade winds. At present, there are few reports on coral skeletal Cd/Ca records in coastal and continental shelf, and most of the existing studies were restricted to annual-decadal variations, which seriously hinders the in-depth study of the geochemical behavior of Cd. In this paper, we used a monthly resolution coral skeletal Cd/Ca record, obtained from the Weizhou Island (WZI) in the northern of South China Sea (SCS), to analyze the variation mode and potential influencing factors of Cd in surface seawater, and to discuss the feasibility of coral skeletal Cd/Ca record as an alternative indicator of sea surface winds.

2. Materials and Methods

2.1. Study Area

Weizhou Island is a volcanic island located in the north SCS (Figure 1), isolated from the Asian continent. The island resides in the East Asian monsoon realm but non-upwelling areas with subtropical maritime climate. WZI has suitable surface seawater temperature (SST, from ~16°C to ~32°C, average of ~25.0°C), surface seawater salinity (SSS, average of ~32.9‰), and annual precipitation (~1,423 mm) for coral reefs. Thus, there are a large number of corals, including *Porites*, distributed here with adequate light, excellent water exchange conditions, and smooth coastal slopes. During the wet season (May–October), the southwest summer monsoon and occasional typhoons bring more than 84% of annual precipitation (Figure S1 in Supporting Information S1). In the dry season (November–April), the prevailing winter monsoon brings cold and dry air masses from the Siberian High. Meanwhile, as a famous tourist resort, the local government pays great attention to the protection and monitoring





Figure 1. Location map of the coral sampling site and study area. The black rectangle indicates Weizhou Island (WZI). The black triangle indicates the sample site (W3 and WZI) used in this study.

of marine environment, thus human activities have been clearly recorded in the statistical yearbook or the bulletin of the environment.

2.2. Analytical Methods

The *Porites* coral was taken from the northwest of WZI (W3: $21^{\circ}4'7''N$, $109^{\circ}5'24''E$, as shown in Figure 1) in October 2015, at a sampling depth of 4 m. In the laboratory, the collected coral samples were cleaned and then dried, cutting into a plate (approximately 8 cm wide and 1 cm thick) along the growth axis. After that, we used X-ray to take photos of the interlaced light and dark density bands. To remove surface organic matter and contaminants, all the samples were soaked in 10% H₂O₂ for 48 hr, then cleaned with Milli-Q water (3 times), and finally air-dried for 48 hr at 60°C. Details of sampling and age models can be referred to our previous research (Xu et al., 2018).

After the pretreatment of the coral samples, a total of 416 powder samples with ~ 3 mg in each, were extracted along the coral growth axis, dissolved with 2% HNO₃ at a sample: solution ratio of 1:6,000, and finally liquid samples were obtained for instrumental determination of the macronutrient and trace element contents. In order to correct the matrix effect of instrument drift, internal standard elements (¹⁰³Rh, ¹¹⁵In, and ¹⁸⁷Re) were added to all samples, and the concentration was 6 ppb. In addition, national geochemical standard materials GBW07129, GBW07133, and GBW07135 were selected as external standards. Cd and Ca contents of all samples were determined by the Thermo Fisher inductively coupled plasma-mass spectrometer (ICP-MS).

Oxygen isotopes were analyzed by Finnigan MAT-253 stable isotope mass spectrometer. Details of elements and isotopes can be referred to the previous researches of Xu et al. (2018) and Jiang et al. (2020).

In this study, assuming consistent intra-annual growth rates of the coral, each annual cycle was determined based on the maximum δ^{18} O values (the lowest temperature) and minimum δ^{18} O values (the highest temperature) points, and the coral skeletal data were processed to monthly resolution by linear interpolation. However, due to the complex biological structure of corals and the multivariate changes in surrounding environment, the actual intra-annual growth rate of corals cannot be consistent. Therefore, this method may result in a certain deviation in the time series of monthly resolution, which requires extra attention in data analysis, so that we use 12-month sliding-window correlation to analysis the data that can effectively eliminate the effect of interference. This method takes 12-month data for local averaging, continued calculation on an interval-by-interval basis, and finally makes the correlation analysis results more credible. The relevant equation as follows (N = 12):

$$\overline{x} = \frac{X_{n-(N-1)} + X_{n-(N-2)} + \dots + X_{n-1} + X_n}{N}$$
$$\overline{y} = \frac{y_{n-(N-1)} + Y_{n-(N-2)} + \dots + Y_{n-1} + Y_n}{N}$$
$$\text{Correl}(X, Y) = \frac{\sum (x - \bar{x})(y - \bar{y})}{\sqrt{\sum (x - \bar{x})^2 \sum (y - \bar{y})^2}}$$

According to our previous research (Wu et al., 2022), the coral Ni/Ca ratios during the P2 period (1994–2008 CE) were relatively low and stable, which is lower than worldwide average (~8.2 nmol/kg). Before 1994, exploration, production, and transportation in oilfields near the WZI had a great impact on the heavy metal contents; after 2008 CE, there were several oil spills in the region, with the development and operation of the Beibu Gulf Economic Zone. Thus, we chose the decade (1999–2008 CE) that is not affected by oil extraction or other representative anthropogenic activities around WZI, with a more pronounced trend as the focus of this paper.



Geochemistry, Geophysics, Geosystems



Figure 2. Comparison of surface seawater temperature, δ^{18} O and Cd/Ca ratios in *Porites lutea* coral skeletons from 1999 to 2008. Data for the dry and wet seasons are divided by two background colors (light yellow and light blue). Coral skeleton Cd/ Ca ratios and δ^{18} O are shown as mean \pm standard deviation (SD), where the dark blue solid line presents mean and the dark blue broken lines indicate mean \pm SD. The orange shaded area is the annual mean of coral Cd/Ca ratios from 1999 to 2008 CE.

2.3. Climate and Environmental Data

The SST data from $0.25^{\circ} \times 0.25^{\circ}$ ERA5 monthly mean data (Hersbach et al., 2019) were utilized in this study (https://cds.climate.copernicus.eu/cdsapp#!/dataset/10.24381/cds.f17050d7?tab = overview/). The $0.5^{\circ} \times 0.5^{\circ}$ SSS data were from the Simple Ocean Data Assimilation (SODA, http://apdrc.soest.hawaii.edu/las/v6/constrain?var=4785/). Rainfall and wind speed data were collected from the WZI Weather Station. The $0.1^{\circ} \times 0.1^{\circ}$ monthly average Chlorophyll-a (Chl-a) concentration data were acquired using NOAA database (https://coastwatch.pfeg.noaa.gov/erddap/griddap/jplMURSST41.html). The East Asian summer/winter monsoon indexes were obtained from the National Tibetan Plateau Data Center (http://data.tpdc.ac.cn/zh-hans/data/1c0c4197-5e5d-4f0d-bd38-03dae3658a06/) and China Climate Bulletin.

3. Results

As shown in Figure 2, coral skeletal Cd/Ca ratios ranged from 4.85 to 45.58 nmol/mol, with an average value of 19.70 ± 9.11 nmol/mol, which is consistent with the range of Cd/Ca ratios in different species of corals (6.53~62.90 nmol/mol) (Pretet et al., 2014). It is worth noting that obvious seasonal fluctuations can be observed with the peaks occurring in the dry season and the low values occurring in the wet season. The average Cd/Ca ratios in the dry season are 23.81 ± 4.81 nmol/mol, and the average Cd/Ca ratios in the wet season are 15.59 ± 6.13 nmol/mol. In addition, the annual average values during the decade ranged from 13.4 to 25.1 nmol/mol, showing a slight upward trend overall (Figure 2).

Cd contents in surface seawater are usually in the range of 0.002–0.909 μ g/L all over the world (Li et al., 2010, 2013; Lian et al., 2001; Yang et al., 2017; Yu, 2003). According to the distribution coefficient of Cd between *Porites lutea* coral skeletons and surface seawater (0.6, referring to Jiang et al. (2020)), the Cd content in surface seawater can be calculated combined with the relatively stable content of Ca in surface seawater (~10.3 mmol/kg) (Kelly et al., 2009; Quinby-Hunt & Turekian, 1983). Results showed that the Cd contents in surface seawater ranged from 0.009 to 0.088 μ g/L, with an average value of 0.038 \pm 0.018 μ g/L (Figure 3). Obviously, the Cd content in surface seawater from WZI was close to the natural seawater, although it was slightly higher than the background value of surface seawater in the northern of SCS (0.002–0.017 μ g/L, Yu, 2003) and the measured value in the open ocean (e.g., 0.04 μ g/L in Xisha Island, Zhou et al., 2007). Comparing with the content of Cd in the near-shore surface seawater (e.g., 0.14 μ g/L in Daya Bay, Yang et al., 2017), it was still relatively low. Meanwhile,





Figure 3. Plot showing Cd content in surface seawater calculated from the Cd/Ca ratios of W3 coral. The dark blue solid line presents mean and the dark blue broken lines indicate mean \pm SD. The blue background indicates the background value of surface seawater in the northern of South China Sea. The gray background indicates the surface seawater measurements from Weizhou Island (data from Yang et al. (2017)).

it is much lower than the national standard of first-class sea water quality. Therefore, it can be inferred that the surface seawater around WZI is not significantly affected by human activities during 1999–2008 CE.

4. Discussion

4.1. Source of Cd in Surface Seawater

The ocean is an important sink for most trace metals (Sabarathinam et al., 2019). According to previous researches, Cd in surface seawater is mainly imported by runoff (Goldstein & Jacobsen, 1988), atmospheric input (Duce & Hoffman, 1976; Marx et al., 2005; Mason, 2013), and remobilization from sediments (Choi et al., 2006; Ni et al., 2016; Sakellari et al., 2011; Simpson, 1978). Moreover, abnormal content variations are usually associated with natural events (e.g., volcanic eruptions) and pollutant inputs (Sun et al., 2016).

4.1.1. Runoff

Terrestrial input, especially runoff, is the primary access to the sea for trace metals (Feng & Tian, 2018; Wang et al., 2010), usually affected by river input and precipitation. Cd mainly exists in zinc, copper and aluminum ores, so it can be transferred from land to the ocean through mining and weathering (Ye et al., 2005). Moreover, the erosion and transport of terrestrial materials on tropical islands occur frequently in high temperature and rainy seasons (Saha et al., 2019). More than 84% of annual precipitation (1999~2008 CE) in the WZI is concentrated from May to October (Figure S1 in Supporting Information S1), which usually carries a large amount of land-based Cd to coastal seawater. However, the peaks of skeletal Cd/Ca ratios all occur in the dry season with poor precipitation (Figure S2 in Supporting Information S1). Therefore, runoff associated with precipitation cannot be responsible for the temporal distribution of Cd in coral skeletons. In addition, WZI is far away from the mainland (~24 nautical miles) and lacks direct input from rivers. Thus, the river source of Cd near the WZI can be ignored. This conclusion is consistent with Li et al. (2019) regarding the sources of rare earth elements in the WZI.

4.1.2. Atmospheric Input

Atmospheric dry/wet deposition is another important source for Cd in surface seawater. Accompanied by wind or precipitation, solid particles usually fall into the sea and dissolve/release trace elements in the sinking process (Libes, 2009). According to Wang et al. (2017), the mean content of Cd in atmospheric input in China (1995~2015) exceeded the primary standard of Environmental Quality Standard for Soils (GB15618-1995) by

15252027, 2022, 11, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GC010600 by Guangxi University, Wiley Online Library on [03/01/2023]. See the Terms

and Con

Geochemistry, Geophysics, Geosystems

10.1029/2022GC010600



Figure 4. Comparison of surrounding environmental parameters and the coral skeletal Cd/Ca ratios in the Weizhou Island (WZI) from 1999 to 2008. This figure shows the relationship between coral Cd/Ca ratios, maximum wind speed, average wind speed, precipitation, surface seawater salinity (SSS) and surface seawater temperature (SST) in the WZI from 1999 to 2008. Data for the dry and wet seasons are divided by two background colors (light yellow and light blue). The blue dotted lines indicate tropical cyclones around WZI (wind speed data from the Weizhou Island Weather Station; SST data from ERA5; SSS data from Simple Ocean Data Assimilation).

16.5 times. It is worth noting that there is more precipitation in the South China, so wet deposition is the main atmospheric source around the WZI. In addition, previous studies have shown that wet depositions can account for 80%–90% of atmospheric input flux of Cd in the oceans around the world (Duce et al., 1991; Patterson & Duce, 1991). However, our data suggested that the precipitation and the Cd/Ca ratios are negatively correlated (r = -0.32, p < 0.05, Figure 4). Meanwhile, the Cd/Ca ratios in the dry season were much higher than those in the wet season during most years (Figure 5), ruling out the possibility of contribution from the atmospheric wet depositions. Therefore, we believed that the atmospheric wet deposition was not the main source of Cd around the WZI. This conclusion is consistent with Duce and Hoffman (1976), which indicated that the contribution of this process to the Cd in the ocean was very limited, although a small amount of land-based materials could be carried to the open ocean through the atmosphere transport. On the other hand, the dry deposition process is affected by the wind forces. As shown in Figure 4, the average wind speeds and coral skeletal Cd/Ca ratios have a similar trend. However, the contents of ²¹⁰Pb in the surface water of the SCS, as a tracer of atmospheric inputs, have been reported extremely low (Nozaki et al., 1998). Therefore, we considered that the dust carried by the winds contributed little, but the similar trend between wind speeds and Cd/Ca ratios suggested that winds, especially winter monsoons, made a significant contribution to the Cd levels in surface seawater around the WZI.

4.1.3. Seawater Vertical Mixing and Sediment Remobilization

Coral skeletons can record the fluctuations of Cd contents in surface seawater during the upwelling process, when deep water is lifted to the surface (Shen et al., 1987, 1992). Thus, the increase of coral skeletal Cd/Ca ratios are often associated with vertical exchange between warm nutrient-limited surface waters and cold nutrient-enriched bottom waters (Lea et al., 1989). In the northern SCS, upwelling is a regular summer phenomenon caused by the southwest monsoon (Chen et al., 2011), while the peaks of Cd/Ca ratios mainly occurred in winter and the study area is relatively shallow without distinct upwelling. Therefore, the seasonal changes of the skeletal Cd/Ca ratios were unlikely to be caused by the upwelling.





Figure 5. Time series highlighting annual changes in coral skeletal Cd/Ca ratios in Weizhou Island (WZI) (1999~2008) and precipitation in dry and wet seasons. This picture shows the comparison of Cd/Ca ratio and precipitation in the WZI from 1999 to 2008. The black-bordered light blue and dark blue column bars represent annual averages of Cd/Ca ratios in the dry and wet seasons, respectively, while the orange-bordered light blue and dark blue column bars represent annual averages of precipitation in the dry and wet seasons, respectively. The yellow background indicates periods of inconsistent trends.

The direct input of dissolved Cd from rivers to coastal seawater is very limited, but the northern SCS receives a large amount of sediments transported by surrounding rivers and accumulates in the continental shelf (Liu et al., 2016), including Pearl River (84 tons/year), Red River (130 tons/year) and Mekong River (160 tons/year) (Cai et al., 2011). The fluvial input of dissolved metals has a very limited impact on the ocean, but those sediments can be transferred to the continental shelf (Rea & Ruff, 1996). Therefore, shelf sediments are also considered to be an potential significant source of dissolved trace metals in the ocean (Souri et al., 2020). Due to the obvious seasonal changes and the prevailing monsoons around WZI, we compared the monthly average wind speeds with the coral skeletal Cd/Ca ratios (Figure 6). It was worth noting that there was a high agreement between the multi-year monthly average wind speeds and coral skeletal Cd/Ca ratios (Figure 6, r = 0.83, p < 0.05). To eliminate effects of deviations in the interpolation time series, we used a 12-month sliding-window correlation analysis testing the strength and stationarity of the relationship between monthly average wind speeds and the monthly coral skeletal Cd/Ca ratios from 1999 to 2008 CE (Figure 7). As shown in Figure 7, the average wind speeds and coral skeletal Cd/Ca ratios were significantly correlated because most sliding correlation coefficients were >0.3 (p < 0.01). Thus, we considered that the seasonal variations of Cd in surface seawater are related to the vertical mixing of coastal seawater induced by winds, which might lead to the re-release of particulate adsorbed Cd from sediments into seawater.

At the bottom of the ocean, Cd can be divided into dissolved phase, particulate phase and sedimentary phase (Zhang et al., 2018). Affected by reductant-oxidant conditions (Jiann & Ho, 2014), Cd content is usually higher in sediments and particulates, and can be reversibly adsorbed on iron and manganese oxides. The interaction between ions is enhanced, leading to the reversal of the adsorption process, and part of the combined Cd can be transformed into free Cd when surrounding salinity increases (Guinoiseau et al., 2018; Jiann & Ho, 2014; Lao et al., 2019). As shown in Figure 4, the SSS was lower in the wet season but higher in the dry season. Therefore, the regular seasonal variation of Cd in surface seawater should be related with the wind speed: in summer, the wind speeds dropped, which conduced to the seawater stratification; in winter, the wind speeds rose, which significantly promoted the vertical mixing process (Tang et al., 2003). This suggested that the processes of seawater vertical mixing and sediment remobilization caused by winds were the main reasons for the seasonal variations of Cd contents in surface seawater. In addition, we noticed that the multi-year monthly average wind speeds





Figure 6. Comparison of the annual trend of coral Cd/Ca ratio, average wind speed and Chl-a in each month from 1999 to 2008 in the Weizhou Island.

showed a significant increase in June–July (Figure 6). Considering it only occurred in individual years from 1999 to 2008 CE (Figure 4), it was likely to be related to tropical cyclones.

4.2. Biogeochemical Process

In winter, the strong winter monsoon in the WZI brings a large amount of nutrients to the surface seawater, which is conducive to the growth and reproduction of phytoplankton and produces many bio-particles (Ma et al., 2014).







Geochemistry, Geophysics, Geosystems



Figure 8. Comparison of the coral surface seawater temperature (SST), Chl-a and Cd/Ca ratios in the Weizhou Island (WZI) from 1999 to 2008. This figure shows the relationship between coral SST, Chl-a and Cd/Ca ratios in the WZI from 1999 to 2008. Two background colors (light yellow and light blue) are used to make a distinction between the dry and wet season (SST data from ERA5; Chl-a data from NOAA).

Chl-a concentration is an important indicator of the biomass (Cai et al., 2002), reflecting eutrophication in the water body (Liu & Zhao, 2019; Oh & Suh, 2006). We found that Chl-a contents in the WZI were relatively high in winter, but began to decrease after spring as the northeast monsoons subsided (Figure 8), and were significantly positive correlated with Cd/Ca ratios (r = 0.50, p < 0.05). Therefore, we believed that the seasonal changes of Chl-a and SST revealed an event of phytoplankton bloom during the winter. In general, Chl-a contents are higher in summer and lower in winter (Wang et al., 2016), and should be negatively correlated with Cd/Ca ratios, due to the fact that an increase in biomass leads to a decrease in dissolved Cd content in surface seawater (Bruland, 1980; Guinoiseau et al., 2018). However, the situation in our research is apparently opposite, suggesting that the reduction in Cd contents in surface seawater due to biological processes might be masked by the impacts from seawater vertical mixing and sediment remobilization caused by surface winds. In addition, the enhanced phytoplankton leads to the forming excess organic matter, which sinks to the bottom and decomposes rapidly (Bruland et al., 1994). This process requires the consumption of dissolved oxygen, but the accelerated movement of the seawater under the action of winter winds weakens the effect of organic matter decomposition. The uptake of Cd by phytoplankton is directly proportional to dissolved Cd concentrations (Sunda & Huntsman, 2000), while the nutrient-rich bottom seawater transported to the surface by winter monsoon, creating a positive feedback process, but gradually subsides as the winter passes. In some years, the peak of Chl-a also appeared in summer, but due to the weakening of seawater mixing, the dissolved oxygen consumed by organic matter degradation cannot be replenished in time, resulting in the state of hypoxia of the seawater which is not conducive for the dissolution of particle Cd. It also confirmed that winter monsoons were strong enough to transport nutrients in bottom water to the surface more efficiently in the study area. In the contrary, the summer monsoons were too weak to have a marked impact on both seawater vertical mixing and sediment remobilization.

4.3. Coral Cd/Ca Ratio as an Indicator of Surface Wind

Coral skeletal Cd/Ca ratios have been successfully applied as a tracer of wind-driven coastal upwelling (Delgadillo-Hinojosa et al., 2015; Geen, 1996), but there is no upwelling near the WZI. The perennial monsoons in the WZI have obvious seasonal variability, characterized by northeast monsoon in the dry season and southwest monsoon in the wet season. During the winter, surface seawater driven by winter monsoon moves, and the lower layer of Cd-rich seawater produces the compensation lift to replenish the surface seawater loss (Reissmann et al., 2009). Meanwhile, the seawater vertical mixing increases the oxygen content of the underlying water to induce sediment remobilization of Cd (Reissmann et al., 2009). In contrast, this effect is limited due to the relatively weak summer monsoon during the summer.





Figure 9. Comparison of the East Asian summer/winter monsoon indexes and coral skeletal Cd/Ca ratios in the Weizhou Island from 1999 to 2008 CE. This figure shows the annual average of Cd/Ca ratios in dry and wet seasons and compared with annual monsoon indexes. Depending on the occurrence time of monsoon, we used summer monsoon indexes from 1999 to 2008 CE and winter monsoon indexes from 1998 to 2007 CE. The East Asian summer/winter monsoon indexes were obtained from Huang and Zhao (2019), Zhao et al. (2015) and China Climate Bulletin.

In present study, the wind speed data were used to present that the winds in the dry season were much higher than those in the wet season, which can give expression to the key role of winter monsoon. Although the site data we used can reflect the correlation, it can not fully reflect the wind intensity of the whole region. Thus, we introduced the Asian summer/winter monsoon indexes, which can present the wind strengths of the entire region, in order to explain the increase of Cd/Ca ratios with no concurrent shift in wind speed. It can be seen from Figure 4 that Cd/Ca ratios increased in summer since 2003, while this change in winter significantly occurred in 2005. The increase in Cd/Ca ratios in the wet/dry season can respond to the strengthen of Asian summer/winter monsoon indexes (Figure 9). During 2003-2008 CE, the Asian summer monsoon indexes trend is similar to Cd/Ca ratios, higher than those during 1999–2002 CE. Likewise, the Asian winter monsoon indexes during 2005–2008 CE were higher than those during 1999-2004 CE. We believed that the strengthen of Asian monsoons might be responsible to the increase in coral Cd/Ca ratios. Comparing the East Asian summer/winter monsoon indexes with the coral skeletal Cd/Ca ratios from 1999 to 2008 CE (Figure 9), we found that the summer monsoon was very weak and had no correlation with the summer average of Cd/Ca ratios (r = 0.15, p > 0.05), while the winter monsoon was strong and had significant correlation with the winter average of Cd/Ca ratios (r = 0.64, p < 0.05). Thus, we can explain the different patterns of coral skeletal Cd/Ca ratios in the dry and wet seasons in 2003 and 2008 CE (Figure 5), relating to the weakening winter monsoon or the strengthening summer monsoon (Figure 9).

Tropical cyclone is also associated with wind speed in the wet season besides monsoons. The coral skeletal Cd/ Ca ratios showed a significant correlation (r = 0.45, p < 0.05) with the maximum wind speeds in the summer. However, the peaks of coral skeletal Cd/Ca ratios were not corresponding to the typhoons affecting WZI from 1999 to 2008 CE (Figure 4), indicating that tropical cyclones contributed limited to the rise of Cd levels in surface seawater probably due to the short duration of tropical cyclones or geochemical sensitivity of Cd (Simpson, 1978). Therefore, considering the high consistency between coral skeletal Cd/Ca ratios and average wind speeds, we suggested that coral skeletal Cd/Ca ratios in the winter might be a potential novel proxy for winter monsoons.

In order to make our conclusions more convincing, we introduced a coral Ba/Ca data set from W3 and a bimonthly coral Ba/Ca data set (2002–2005 CE) from another *Porites lutea* coral (Figure 1, WZI coral) near WZI reported by Li et al. (2022) as an extended multiproxy approach. Considering the effects on the Ba/Ca ratios from the



15252027, 2022, 11, Downloaded from

elibrary.wiley.com

doi/10.1029/2022GC010600 by Guangxi University

Wiley

Online Library on [03/01/2023]. See the Terms and Condi-

(nups

QA

y une

applicable

high barite (BaSO₄) consumption used for oil exploration and drilling activities before 2004 CE reported by Li et al. (2022), we selected the Ba/Ca time series from 2004 to 2005 CE of WZI coral to compare with the Ba/Ca and Cd/Ca ratios (2004–2008 CE) of W3 coral (Figure S3 in Supporting Information S1). Coincidentally, the Ba/Ca records from two corals both exhibited strong seasonal variations, similar to Cd/Ca ratios. Therefore, we believed that the data set of W3 coral should be representative.

5. Conclusion

This is the first to report the relationship between monthly resolution Cd/Ca ratios of coral and winter monsoons, revealing its great potential for the documentation of prehistoric changes in winter monsoons. In the absence of anthropogenic source, the coral Cd/Ca ratios exhibited obvious seasonal fluctuations, and had significant correlations with the speeds of sea surface winds. Under the influences of strong surface winds, the vertical mixing process can change the redox conditions of bottom seawater, leading to the release of the adsorbed Cd in sediments into seawater. Despite that this process also leads to an increase of nutrients, which promotes the growth of phytoplankton resulting in the depletion of Cd the surface seawater, the reduction of Cd levels in surface seawater connected with biological processes is masked by sediment remobilization caused by winter monsoons. Compared with strong winter monsoons, summer monsoons are always very weak and tropical cyclones occur for a short period of time, indicating a limited effect on Cd contents in surface seawater. Therefore, we considered that the sediment remobilization and mixing processes driven by winter monsoons should play a key role in the seasonal variations of Cd contents in surface seawater, and coral skeletal Cd/Ca ratios in the winter have potentials to be a potential novel proxy for winter monsoons in the non-upwelling shelf environments.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The data used in this paper will be deposited in a general data repository (https://doi.org/10.6084/m9.figshare.21087820.v1).

References

Boyle, E. A. (1988). Cadmium: Chemical tracer of deepwater paleoceanography. *Paleoceanography*, 3(4), 471–489. https://doi.org/10.1029/ PA003i004p00471

Boyle, E. A., Sclater, F., & Edmond, J. M. (1976). On the marine geochemistry of cadmium. *Nature*, 263(5572), 42-44. https://doi.org/10.1038/263042a0

Bruland, K. W. (1980). Oceanographic distributions of cadmium, zinc, nickel, and copper in the North Pacific. Earth and Planetary Science Letters, 47(2), 176–198. https://doi.org/10.1016/0012-821X(80)90035-7

- Bruland, K. W., Knauer, G. A., & Martin, J. N. (1978). Cadmium in northeast Pacific waters. *Limnology & Oceanography*, 25(4), 618–625. https://doi.org/10.4319/10.1978.23.4.0618
- Bruland, K. W., Orians, K. J., & Cowen, J. P. (1994). Reactive trace metals in the stratified central North Pacific. *Geochimica et Cosmochimica Acta*, 58(15), 3171–3182. https://doi.org/10.1016/0016-7037(94)90044-2
- Cai, G., Peng, X., & Zhang, Y. (2011). The significances of and advances in the study of sediment sources in the South China Sea (in Chinese). Advances in Marine Science, 29(1), 113–121.
- Cai, Y., Ning, X., & Liu, C. (2002). Distribution characteristics of size-fractionated chlorophyll a and productivity of phytoplankton in the northern South China Sea and Beibu Gulf during August 1999 (in Chinese). *Studia Marina Sinica*, 44, 11–21.
- Carriquiry, J. D., & Villaescusa, J. A. (2010). Coral Cd/Ca and Mn/Ca records of ENSO variability in the Gulf of California. *Climate of the Past*, 6(1), 401–410. https://doi.org/10.5194/cp-6-401-2010
- Chen, T., Yu, K., Li, S., Chen, T., & Shi, Q. (2011). Anomalous Ba/Ca signals associated with low temperature stresses in *Porites* corals from Daya Bay, northern South China Sea. *Journal of Environmental Sciences*, 23(9), 1452–1459. https://doi.org/10.1016/S1001-0742(10)60606-7 Choi, S. C., Wai, O. W. H., Choi, T. W. H., Li, X. D., & Tsang, C. W. (2006). Distribution of cadmium, chromium, copper, lead and zinc in marine
- sediments in Hong Kong waters. Environmental Geology, 51(3), 455–461. https://doi.org/10.1007/s00254-006-0340-z
- Collier, R., & Edmond, J. (1984). The trace element geochemistry of marine biogenic particulate matter. *Progress in Oceanography*, *13*(2), 113–199. https://doi.org/10.1016/0079-6611(84)90008-9
- Delgadillo-Hinojosa, F., Camacho-Ibar, V., Huerta-Díaz, M. A., Torres-Delgado, V., Pérez-Brunius, P., Lares, L., et al. (2015). Seasonal behavior of dissolved cadmium and Cd/PO₄ ratio in Todos Santos Bay: A retention site of upwelled waters in the Baja California peninsula, Mexico. *Marine Chemistry*, 168, 37–48. https://doi.org/10.1016/j.marchem.2014.10.010
- Duce, R. A., & Hoffma, G. L. (1976). Atmospheric vanadium transport to the ocean. Atmospheric Environment, 10(11), 989–996. https://doi.org/10.1016/0004-6981(76)90207-9

Acknowledgments

This research was supported by the National Natural Science Foundation of China (Grant 42030502, 41976059, and 42090041) and the project supported by Innovation Group Project of Southern Marine Science and Engineering Guangdong Laboratory (Zhuhai) (No. 311019006/311020006). We are grateful to Ning Wang, Haodan Yang, and Xingyuan Wu from Guangxi University for sample collection and testing. Thanks are due to the Editor Adina Paytan and two reviewers for their critical reviews and constructive comments, which helped us improve significantly our paper.

- Duce, R. A., Liss, P. S., Merrill, J. T., Atlas, E. L., Buat-Menard, P., Hicks, B. B., et al. (1991). The atmospheric input of trace species to the world ocean. *Global Biogeochemical Cycles*, 5(3), 193–259. https://doi.org/10.1029/91GB01778
- Feng, Y., & Tian, J. (2018). Hydrochemical behaviors of Cd in paleoceans and a review of Cd/Ca ratio in foraminifera shells as a paleoeanographyic proxy (in Chinese). Marine Geology & Quaternary Geology, 38(5), 122–129. https://doi.org/10.16562/j.cnki.0256-1492.2018.05.012
- Geen, A. (1996). Cadmium in the California Current system: Tracer of and present upwelling. Journal of Geophysical Research, 101(C2), 3489–3507. https://doi.org/10.1029/95JC03302
- Goldstein, S. J., & Jacobsen, S. B. (1988). Rare Earth elements in river waters. Earth and Planetary Science Letters, 89(1), 35–47. https://doi. org/10.1016/0012-821X(88)90031-3
- Grottoli, A. G., Matthews, K. A., Palardy, J. E., & McDonough, W. F. (2013). High resolution coral Cd measurements using LA-ICP-MS and ID-ICP-MS: Calibration and interpretation. *Chemical Geology*, 356, 151–159. https://doi.org/10.1016/j.chemgeo.2013.08.024
- Guinoiseau, D., Galer, S. J. G., & Abouchami, W. (2018). Effect of cadmium sulphide precipitation on the partitioning of Cd isotopes: Implications for the oceanic Cd cycle. *Earth and Planetary Science Letters*, 498(15), 300–308. https://doi.org/10.1016/j.epsl.2018.06.039
- Hersbach, H., Bell, B., Berrisford, P., Biavati, G., & Horányi, A. (2019). ERA5 monthly averaged data on single levels from 1959 to present. Copernicus Climate Change Service (C3S) Climate Data Store (CDS). https://doi.org/10.24381/cds.f17050d7
- Horner, T. J., Lee, R. B. Y., Henderson, G. M., & Rickaby, R. E. M. (2013). Nonspecific uptake and homeostasis drive the oceanic cadmium cycle. Proceedings of the National Academy of Sciences of the United States of America, 110(7), 2500–2505. https://doi.org/10.1073/ pnas.1213857110
- Huang, G., & Zhao, G. (2019). The East Asian summer monsoon index (1851–2021). National Tibetan Plateau Data Center, 2019. https://doi. org/10.11888/Meteoro.tpdc.270323
- Jiang, W., Yu, K., Wang, N., Yang, H., Yang, H., Xu, S., et al. (2020). Distribution coefficients of trace metals between modern coral-lattices and seawater in the northern South China Sea: Species and SST dependencies. *Journal of Asian Earth Sciences*, 187, 1–8. https://doi.org/10.1016/j. jseaes.2019.104082
- Jiann, K.-T., & Ho, P. (2014). Cadmium mixing behavior in estuaries: Redox controls on removal and mobilization. Terrestrial, Atmospheric and Oceanic Sciences, 25(5), 655–664. https://doi.org/10.3319/TAO.2014.04.01.01(Oc)
- Jiann, K.-T., Wen, L.-S., & Santschi, P. H. (2005). Trace metal (Cd, Cu, Ni and Pb) partitioning, affinities and removal in the Danshuei River estuary, a macro-tidal, temporally anoxic estuary in Taiwan. Marine Chemistry, 96(3–4), 293–313. https://doi.org/10.1016/j.marchem.2005.03.001
- Jurado-González, J. A., Galindo-Riaño, M. D., & García-Vargas, M. (2003). Experimental designs in the development of a new method for the sensitive determination of cadmium in seawater by adsorptive cathodic stripping voltammetry. *Analytica Chimica Acta*, 487(2), 229–241. https://doi.org/10.1016/s0003-2670(03)00536-1
- Kelly, A. E., Reuer, M. K., Goodkin, N. F., & Boyle, E. A. (2009). Lead concentrations and isotopes in corals and water near Bermuda, 1780–2000. Earth and Planetary Science Letters, 283(1–4), 93–100. https://doi.org/10.1016/j.epsl.2009.03.045
- Lane, T. W., & Morel, F. M. M. (2000). A biological function for cadmium in marine diatoms. Proceedings of the National Academy of Sciences of the United States of America, 97(9), 4627–4631. https://doi.org/10.1073/pnas.090091397
- Lane, T. W., Saito, M. A., George, G. N., Pickering, I. J., Prince, R. C., & Morel, F. M. M. (2005). A cadmium enzyme from a marine diatom. *Nature*, 435(7032), 42–456. https://doi.org/10.1038/434455a
- Lao, Q., Su, Q., Liu, G., Shen, Y., Chen, F., Lei, X., et al. (2019). Spatial distribution of and historical changes in heavy metals in the surface seawater and sediments of the Beibu Gulf, China, Marine Pollution Bulletin, 146, 427–434, https://doi.org/10.1016/j.marpolbul.2019.06.080
- Lea, D. W., Shen, G. T., & Boyle, E. A. (1989). Coralline barium records temporal variability in equatorial Pacific upwelling. *Nature*, 340(6232), 373–376. https://doi.org/10.1038/340373a0
- Li, J., Li, J., Zheng, L., Wang, Z., Chen, F., & Wang, X. (2013). Distribution of heavy metals in the mid-South part of South China Sea Water (in Chinese). Environmental Monitoring in China, 29(3), 65–71. https://doi.org/10.19316/j.issn.1002-6002.2013.03.013
- Li, X., Liu, Y., Wu, C.-C., Sun, R., Zheng, L., Lone, M. A., & Shen, C. C. (2019). Coral-inferred monsoon and biologically driven fractionation of offshore seawater rare Earth elements in Beibu Gulf, northern South China Sea. Solid Earth Sciences, 4(4), 131–141. https://doi.org/10.1016/j. sesci.2019.09.003
- Li, X., Zhang, L., Liu, Y., & Sun, W. (2022). The Ba/Ca record of coral from Weizhou Island: Contributions from oil-drilling muds and the winter monsoon. *Marine Pollution Bulletin*, 174, 113317. https://doi.org/10.1016/j.marpolbul.2021.113317
- Li, Y., Tan, L., & Wang, J. (2010). The temporal and spatial distribution of cadmium Plumbum Mercury arsenic of Shandong southern coastal surface Sea Water (in Chinese). Periodical of Ocean University of China, 40(S1), 179–184. https://doi.org/10.16441/j.cnki.hdxb.2010.s1.029
- Lian, X., Wang, Y., & Chen, Q. (2001). Assessment on heavy metals in seawater, surface sediments and organisms at Guangxi inshore (in Chinese). *Marine Environmental Science*, 20(2), 59–62.
- Libes, S. (2009). Introduction to marine biogeochemistry. Academic Press.
- Lionetto, M. G., Caricato, R., Erroi, E., Giordano, M. E., & Schettino, T. (2005). Carbonic anhydrase-based environmental bioassay. International Journal of Environmental Analytical Chemistry, 85(12–13), 895–903. https://doi.org/10.1080/03067310500154320
- Liu, D., & Zhao, Q. (2019). Study on the spatial and temporal distribution of chlorophyll a concentration in Beibu Gulf (in Chinese). Journal of Marine Sciences, 37(2), 95–102. https://doi.org/10.3969/j.issn.1001-909X.2019.02.010
- Liu, X., Lou, C., Xu, L., & Sun, L. (2012). Distribution and bioavailability of cadmium in ornithogenic coral-sand sediments of the Xisha archipelago, South China Sea. *Environmental Pollution*, 168, 151–160. https://doi.org/10.1016/j.envpol.2012.04.018
- Liu, Z., Zhao, Y., Colin, C., Stattegger, K., Wiesner, M. G., Huh, C.-A., et al. (2016). Source-to-sink transport processes of fluvial sediments in the South China Sea. *Earth-Science Reviews*, 153(1), 238–273. https://doi.org/10.1016/j.earscirev.2015.08.005
- Ma, W., Chai, F., Xiu, P., Xue, H., & Tian, J. (2014). Simulation of export production and biological pump structure in the South China Sea. Geo-Marine Letters, 34(6), 541–554. https://doi.org/10.1007/s00367-014-0384-0
- Martin, J. H., & Knauer, G. A. (1973). The elemental composition of plankton. Geochimica et Cosmochimica Acta, 37(7), 1639–1653. https:// doi.org/10.1016/0016-7037(73)90154-3
- Marx, S. K., Kamber, B. S., & McGowan, H. A. (2005). Provenance of long-travelled dust determined with ultra-trace-element composition: A pilot study with samples from New Zealand glaciers. *Earth Surface Processes and Landforms*, 30(6), 699–716. https://doi.org/10.1002/ esp.1169
- Mason, R. P. (2013). Metal(loid)s in the atmosphere and their inputs to surface waters. In Trace metals in aquatic systems (pp. 167-218).
- Ni, L., Li, D., Su, L., Xu, J., Li, S., Ye, X., et al. (2016). Effects of algae growth on cadmium remobilization and ecological risk in sediments of Taihu Lake. *Chemosphere*, 151, 37–44. https://doi.org/10.1016/j.chemosphere.2016.02.012
- Nozaki, Y., Dobashi, F., Kato, Y., & Yamamoto, Y. (1998). Distribution of Ra isotopes and the ²¹⁰Pb and ²¹⁰Po balance in surface seawaters of the mid Northern Hemisphere. *Deep Sea Research Part I: Oceanographic Research Papers*, 45(8), 1263–1284. https://doi.org/10.1016/S0967-0637(98)00016-8

- Oh, H.-J., & Suh, Y.-S. (2006). Temporal and spatial characteristics of chlorophyll a distributions related to the oceanographic conditions in the Korean waters. Journal of the Korean Association of Geographic Information Studies, 9(3), 36–45.
- Patterson, T. L., & Duce, R. A. (1991). The cycle of atmospheric cadmium over the North Pacific Ocean. Tellus B: Chemical and Physical Meteorology, 43(1), 12–29. https://doi.org/10.1034/j.1600-0889.1991.00002.x
- Pretet, C., Reynaud, S., Ferrier-Pagès, C., Gattuso, J.-P., Kamber, B. S., & Samankassou, E. (2014). Effect of salinity on the skeletal chemistry of cultured scleractinian zooxanthellate corals: Cd/Ca ratio as a potential proxy for salinity reconstruction. *Coral Reefs*, 33(1), 169–180. https:// doi.org/10.1007/s00338-013-1098-x
- Quinby-Hunt, M. S., & Turekian, K. K. (1983). Distribution of elements in Sea Water. EOS Transactions, 64(14), 130–132. https://doi. org/10.1029/EO064i014p00130
- Rea, D. K., & Ruff, L. J. (1996). Composition and mass flux of sediment entering the world's subduction zones: Implications for global sediment budgets, great earthquakes, and volcanism. Earth and Planetary Science Letters, 140(1–4), 1–12. https://doi.org/10.1016/0012-821X(96)00036-2
- Reissmann, J. H., Burchard, H., Feistel, R., Hagen, E., Lass, H. U., Mohrholz, V., et al. (2009). Vertical mixing in the Baltic Sea and consequences for eutrophication—A review. *Progress in Oceanography*, 82(1), 47–80. https://doi.org/10.1016/j.pocean.2007.10.004
- Reuer, M. K., Boyle, E. A., & Cole, J. E. (2003). A mid-twentieth century reduction in tropical upwelling inferred from coralline trace element proxies. *Earth and Planetary Science Letters*, 210(3), 437–452. https://doi.org/10.1016/S0012-821X(03)00162-6
- Sabarathinam, C., Bhandary, H., & Al-Khalid, A. (2019). A geochemical analogy between the metal sources in Kuwait Bay and territorial sea water of Kuwait. *Environmental Monitoring and Assessment*, 191(3), 1–19. https://doi.org/10.1007/s10661-019-7219-4
- Saha, N., Webb, G. E., Christy, A. G., & Zhao, J. (2019). Vanadium in the massive coral Porites: A potential proxy for historical wood clearing and burning. *Earth and Planetary Science Letters*, 527(1), 1–12. https://doi.org/10.1016/j.epsl.2019.115793
- Saha, N., Webb, G. E., & Zhao, J. (2016). Coral skeletal geochemistry as a monitor of inshore water quality. Science of the Total Environment, 566–567, 652–684. https://doi.org/10.1016/j.scitotenv.2016.05.066
- Sakellari, A., Plavšić, M., Karavoltsos, S., Dassenakis, M., & Scoullos, M. (2011). Assessment of copper, cadmium and zinc remobilization in Mediterranean marine coastal sediments. *Estuarine, Coastal and Shelf Science*, 91, 1–12. https://doi.org/10.1016/j.ecss.2010.09.008
- Shen, G. T., Boyle, E. A., & Lea, D. W. (1987). Cadmium in corals as a tracer of historical upwelling and industrial fallout. *Nature*, 328(6133), 794–796. https://doi.org/10.1038/328794a0
- Shen, G. T., Cole, J. E., Lea, D. W., Linn, L. J., McConnaughey, T. A., & Fairbanks, R. G. (1992). Surface ocean variability at Galapagos from 1936–1982: Calibration of geochemical tracers in corals. *Paleoceanography and Paleoclimatology*, 7(5), 563–588. https://doi. org/10.1029/92PA01825
- Simpson, W. R. (1978). A critical review of cadmium in the marine environment. Institute of Oceanographic Sciences.
- Souri, A., Masoodi, M., Niyogi, S., & Naji, A. (2020). Speciation and risk assessment of selected trace metals in bottom sediment of coral reef ecosystems of the Persian Gulf. *Environmental Science and Pollution Research*, 27(17), 21416–21428. https://doi.org/10.1007/ s11356-020-08632-4
- Sun, R., Hintelmann, H., Liu, Y., Li, X., & Dimock, B. (2016). Two centuries of coral skeletons from the northern South China Sea record Mercury emissions from modern Chinese wars. Environmental Science and Technology, 50(11), 5481–5488. https://doi.org/10.1021/acs.est.5b05965
- Sunda, W. G., & Huntsman, S. A. (2000). Effect of Zn, Mn, and Fe on Cd accumulation in phytoplankton: Implications for oceanic Cd cycling. Limnology & Oceanography, 45(7), 1501–1516. https://doi.org/10.4319/lo.2000.45.7.1501
- Tang, D., Kawamura, H., Lee, M.-A., & Dien, T. V. (2003). Seasonal and spatial distribution of chlorophyll-a concentrations and water conditions in the Gulf of Tonkin, South China Sea. *Remote Sensing of Environment*, 85(4), 475–483. https://doi.org/10.1016/S0034-4257(03)00049-X
- Turner, A., Le Roux, S. M., & Millward, G. E. (2008). Adsorption of cadmium to iron and manganese oxides during estuarine mixing. *Marine Chemistry*, 108(1–2), 77–84. https://doi.org/10.1016/j.marchem.2007.10.004
- Wang, C., Wang, X., Su, R., Li, K., Liang, S., & Yang, S. (2010). The estimation of copper lead zinc and cadmium fluxes into the sea area of the East China Sea interferenced by terrigenous matter and their environmental capacities (in Chinese). Acta Oceanologica Sinica, 32(4), 62–75.
- Wang, K., Wu, J., Du, J., & Wang, N. (2016). Distribution of chlorophyll-a concentration in sea waters of Liaodong Bay in summer and winter (in Chinese). Fisheries Science, 35(6), 1003–1111. https://doi.org/10.16378/j.cnki.1003-1111.2016.06.012
- Wang, M., Yuan, M., & Su, D. (2017). Characteristics and spatial-temporal variation of heavy metals in atmospheric dry and wet deposition of China (in Chinese). *China Environmental Science*, 37(11), 4085–4096.
- Wu, X., Jiang, W., Yu, K., Xu, S., Yang, H., Wang, N., et al. (2022). Coral-inferred historical changes of nickel emissions related to industrial and transportation activities in the Beibu Gulf, northern South China Sea. *Journal of Hazardous Materials*, 424(15), 1–9. https://doi.org/10.1016/j. jhazmat.2021.127422
- Xu, S., Yu, K., Wang, Y., Liu, T., Jiang, W., Wang, S., & Chu, M. (2018). Oil spill recorded by skeletal δ¹³C of *Porites* corals in Weizhou island, Beibu Gulf, northern South China Sea. *Estuarine, Coastal and Shelf Science*, 207, 338–344. https://doi.org/10.1016/j.ecss.2018.04.031
- Xu, Y., Feng, L., Jeffrey, P. D., Shi, Y., & Morel, F. M. M. (2008). Structure and metal exchange in the cadmium carbonic anhydrase of marine diatoms. *Nature*, 452(7183), 56–61. https://doi.org/10.1038/nature06636
- Yang, H., Wang, S., Yu, K., Huang, X., Zhang, R., Wang, Y., et al. (2017). Pollution characteristics of heavy metals in seawater of coral growing regions in the Northern South China Sea (in Chinese). *Ecology and Environmental Sciences*, 26(2), 253–260.
- Yang, W., Cao, Z., Zhang, H., & Lang, Y. (2021). A national wide evaluation of heavy metals pollution in surface sediments from different marginal seas along China Mainland. *Regional Studies in Marine Science*, 42, 101637. https://doi.org/10.1016/j.rsma.2021.101637
- Ye, L., Pan, Z., Li, C., Liu, T., & Xia, B. (2005). The present situation and prospects of geochemical researches on cadmium (in Chinese). Acta Petrologica et Mineralogica, 24(4), 339–348.
- Yu, T. (2003). Study about dissolved heavy metals and their background values in South China Sea (in Chinese). Journal of Oceanography in Taiwan Strait, 22(3), 329–333.
- Zhang, J., Zhou, F., Chen, C., Sun, X., Shi, Y., Zhao, H., et al. (2018). Spatial distribution and correlation characteristics of heavy metals in the seawater, suspended particulate matter and sediments in Zhanjiang Bay, China. PLoS One, 13(8), e0201414. https://doi.org/10.1371/journal. pone.0201414
- Zhang, M., He, P., Qiao, G., Huang, J., Yuan, X., & Li, Q. (2019). Heavy metal contamination assessment of surface sediments of the Subei Shoal, China: Spatial distribution, source apportionment and ecological risk. *Chemosphere*, 223, 211–222. https://doi.org/10.1016/j. chemosphere.2019.02.058
- Zhao, G. J., Huang, G., Wu, R. G., Tao, W. C., Gong, H. N., Qu, X., & Hu, K. M. (2015). A new upper-level circulation index for the East Asian summer monsoon variability. *Journal of Climate*, 28(24), 9977–9996. https://doi.org/10.1175/JCLI-D-15-0272.1
- Zhou, J., Yang, D., Peng, Z., Song, S., Zhou, W., He, J., et al. (2007). The concentrations of dissolved heavy metals in Xisha waters and their influential factors (in Chinese). Journal of University of Science and Technology of China, 37(8), 1036–1042.